

Journal of Geophysical Research: Oceans

RESEARCH ARTICLE

10.1002/2017JC013460

Key Points:

- Pan-Arctic modeling of advective pathways is necessary to understand the long-term (2 years) fate of pollutants spilled in the Arctic Ocean
- Long-term fate of pollutants dependent on where the spill occurs: the Northern Sea Route can be split into three main regimes
- Subduction into deep water could render pollutants irrecoverable. This is a risk especially for spills occurring in the Barents Sea

Correspondence to:

S. Kelly, s.kelly@soton.ac.uk

Citation:

Kelly, S., Popova, E., Aksenov, Y., Marsh, R., & Yool, A. (2018). Lagrangian modeling of Arctic Ocean circulation pathways: Impact of advection on spread of pollutants. *Journal of Geophysical Research: Oceans*, *123*, 2882–2902. https://doi.org/10.1002/ 2017JC013460

Received 5 OCT 2017 Accepted 26 MAR 2018 Accepted article online 6 APR 2018 Published online 25 APR 2018

© 2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Lagrangian Modeling of Arctic Ocean Circulation Pathways: Impact of Advection on Spread of Pollutants

JGR

ഹി

S. Kelly¹ D, E. Popova², Y. Aksenov², R. Marsh¹ D, and A. Yool² D

¹Ocean and Earth Science, University of Southampton, Southampton, UK, ²Marine Systems Modelling, National Oceanography Centre, Southampton, UK

Abstract Sea-ice-free summers are projected to become a prominent feature of the Arctic environment in the coming decades. From a shipping perspective, this means larger areas of open water in the summer, thinner and less compact ice all year round, and longer operating seasons. Therefore, the possibility for easier navigation along trans-Arctic shipping routes arises. The Northern Sea Route (NSR) is one trans-Arctic route, and it offers a potential 10 day shortcut between Western Europe and the Far East. More ships transiting the NSR means an increased risk of an accident, and associated oil spill, occurring. Previous research suggests that current infrastructure is insufficient for increased shipping. Therefore, should an oil spill occur, the window for a successful clean-up will be short. In the event of a failed recovery, the long-term fate of the unrecovered pollutants must be considered, at least until the next melt season when it could become accessible again. Here we investigate the role of oceanic advection in determining the long-term fate of Arctic pollutants using a high-resolution ocean model along with Lagrangian particle-tracking to simulate the spread of pollutants. The resulting "advective footprints" of pollutants are proposed as an informative metric for analyzing such experiments. We characterize the circulation along different parts of the NSR, defining three main regions in the Eurasian Arctic, and relate the distinctive circulation pathways of each to the long-term fate of spilled oil. We conclude that a detailed understanding of ocean circulation is critical for determining the long-term fate of Arctic pollutants.

Plain Language Summary The Earth's climate is changing and the Arctic Ocean is projected to experience ice free summers within decades. This would enable more commercial shipping, which in turn makes an Arctic shipping accident more likely. This could lead to oil (or other pollutants) being spilled into the ocean. Because of the harsh Arctic environment, an oil spill may not be successfully recovered, so we need to consider where it will go in the following months and years. We released virtual "particles" into a computer model of the ocean and tracked their progress for 2 years. In this time, particles traveled, on average, 1,223 km. This demonstrates that pan-Arctic modeling is needed in the event of an unrecovered pollutant spill. Unrecovered oil from one season may be accessible the next spring. By analyzing the spread of our particles, we found that on average 676,917 km² would need to be searched to find it, but that this is highly dependent on where the spill occurs. Finally, we noted that in some places, particularly the Barents Sea, there was a risk that spilled pollutants could become entrained into deep water, rendering them irrecoverable.

1. Introduction

Marine oil spills are a major concern both environmentally and economically. The financial cost of accidental oil spills can run to billions of dollars, and they have the potential to cause significant damage to marine habitats by contaminating the food web and polluting large stretches of coastline (Carson et al., 2003). It is not possible to completely eliminate the risk of an oil spill occurring, and even thorough clean-up operations can leave some unrecovered oil with environmental impacts (Peterson et al., 2003). Therefore, it is important to understand where spilled oil is likely to be transported to, in order to predict the likely environmental, economic, and social consequences of an unrecovered spill.

An oil spill in the Arctic Ocean is becoming increasingly likely. Permanent Arctic sea-ice is retreating rapidly, and it is predicted that the ocean will be seasonally ice-free by the middle of this century if greenhouse emissions continue at their current rate (Boe et al., 2009; Overland & Wang, 2013; Wang & Overland, 2012).



Journal of Geophysical Research: Oceans



Figure 1. (a) The Northern Sea Route (NSR) and release sites for Lagrangian experiments (see section 2.2). There is no fixed definition of the NSR, so we have defined a "main" route via straits (solid line) and a more poleward "alternative" route (dashed). (b) Schematic of Arctic surface circulation. Red: Atlantic inflow following the Arctic Circumpolar Boundary Current (including Barents Sea Branch) and branching at the Lomonosov Ridge. Green: Pacific inflow, following three main pathways: the Alaskan Shelf-break Jet, into the Chukchi Sea through the Herald Canyon, and central flow across the shelf. Note that the flow north of the Canadian Archipelago is currently not definitively established, and presumed flow is shown. (Aksenov et al., 2011). Magenta: Beaufort Gyre. Orange: Transpolar Drift Stream, from Siberia to the Fram Strait.

This "opening up" of the ocean is fuelling increased interest in using the Arctic for commercial shipping (Aksenov et al., 2017; Lee & Kim, 2015). Figure 1a illustrates one such shipping route, the Northern Sea Route (hereafter NSR), which connects the Atlantic gateway to the Arctic with the Bering Strait and the Pacific Ocean. A schematic of this route is presented in Figure 1.

In turn, this increased shipping activity (Østreng et al., 2013) increases the probability of an accidental oil spill from a commercial tanker or from offshore operations occurring. Winter sea-ice will persist in the Arctic for the foreseeable future, however it is likely to be reduced in thickness and extent, and it will be more mobile (Aksenov et al., 2017). Nonetheless, sea-ice remains a considerable risk for potential shipping accidents, and thus for potential oil spills. The harsh Arctic environment and remoteness of the ocean make this a particularly risky place for a spill to occur.

Once released into the ocean, many factors can govern the fate of spilled oil. These depend on the type of oil released, whether dispersants have been applied, and the environment in which the spill occurs (Afenyo et al., 2015; Mariano et al., 2011). Mixing or dissolution into the water column, photo-oxidation, emulsification, evaporation, sedimentation, biodegradation, and ingestion into the food web are all potential fates for spilled oil (Mariano et al., 2011; Xie et al., 2007).

Oil spills in ice-covered areas behave differently to those in other parts of the world due to their interaction with sea-ice (Afenyo et al., 2015). Ordinarily, oil within the water column, or at the ocean's surface, is transported by ocean currents. However, oil can become trapped and corralled in leads between areas of sea-ice, and may even become fully "encapsulated" into growing ice, effectively isolating it from the ocean below. In this manner, oil can be transported by sea-ice and later, upon ice melt, released back to the ocean in a different location, far from the original spill (Afenyo et al., 2015; Izumiyama et al., 2004).

Leaving aside sea-ice interactions, advection due to ocean currents is the dominant process in determining what will happen to oil that has been mixed, dissolved, or otherwise entrained into the water column in the following months and years (Afenyo et al., 2015). Therefore, understanding the advective pathways in the ocean is key for understanding the long-term fate of spilled oil. The other option is that oil could for at least some of its trajectory be transported by sea-ice, however that is beyond the scope of this work.



Biodegradation is a likely eventual fate of an oil spill, but biological processes are inhibited by the freezing temperatures in the Arctic. This means that biodegradation would take longer in the Arctic compared to the rest of the global ocean, meaning that spilled oil would remain an active pollutant for months to years and so long-term consideration of its fate is required (Fingas & Hollebone, 2003). Additionally, due to the short operational season, large distances to ports and other infrastructure, and the generally challenging Arctic environment, there is a significant chance that if a spill occurs, it will not be fully recovered before the winter freezeup makes it inaccessible. During this time, ocean currents and sea-ice can transport oil hundreds of kilometers away from the initial spill location (Main et al., 2016).

Aside from in the Arctic, recent major oil spills have included the Deepwater Horizons incident in the Gulf of Mexico. Ocean models were used to investigate the spread of pollutants here (Macfadyen et al., 2013; North et al., 2013; Weisberg et al., 2017), including Lagrangian analysis of modeled currents. It was found that ocean currents played a dominant role in determining where oil would go, and this held until oil reached the coast, when Stokes drift became the important factor for determining how the oil would beach (Weisberg et al., 2017).

To date, no major spill has occurred in the Arctic. An oil rig ran aground in the Chukchi Sea in 2012, but this did not lead to actual pollution (Meier et al., 2014). However, there has been one particularly notable oil spill in ice-affected waters: In 1989, a major spill from the Exxon Valdez tanker occurred off the coast of Alaska (Peterson et al., 2003). One contributing factor to this event was the tanker deviating from its normal shipping lane to avoid icebergs. This caused an accident, and 11 million gallons (50,000 m³) of Prudhoe Bay crude oil was spilled into the ocean. The financial cost of the spill ran to billions of dollars, with Exxon spending over \$2 billion on oil spill response and restoration (Carson et al., 2003).

Consequently, preparedness modeling around ocean circulation in the Arctic is vital to understand where oil spilled in the region will be located in the springtime when it becomes accessible again, both to permit estimation of the level of the potential recovery costs, and to understand the likely domain and severity of environmental and ecological impacts should a spill similar to Exxon Valdez occur within the Arctic Ocean.

By way of a summary, large-scale circulation pathways in the deep Arctic Ocean are primarily driven by the wind and by the inflows from the North Atlantic and Pacific oceans (Aksenov et al., 2011; Pnyushkov et al., 2013, 2015; Proshutinsky et al., 2015; Proshutinsky & Johnson, 1997). Since the ocean stratification below the Arctic halocline is week, the intermediate depth currents are strongly influenced by the by the oceanic ridges and steep topography of the continental shelf slope (Aksenov et al., 2011). The exchanges between the shelf and the deep part of the ocean occurs through cascading of the dense shelf waters (Ivanov et al., 2015; Ivanov & Golovin, 2007) and the cross-slope currents, driven by the along-shelf component of the wind stress through the Ekman transport mechanism (Bacon et al., 2014).

The presence of sea-ice serves to decrease wind forcing and makes the ocean circulation relatively slow, however this "inhibition" is anticipated to decline as the ice retreats. In bathymetric terms, the Arctic Ocean is roughly evenly divided between shelf seas (up to 200 m depth) and deep ocean, with the latter split by the Lomonosov Ridge into the Amerasian and Eurasian basins (Björk et al., 2007).

As Figure 1b illustrates, the anticyclonic Beaufort Gyre (magenta) dominates the circulation in the Amerasian Basin (Bluhm et al., 2015; Proshutinsky & Johnson, 1997). Meanwhile, cyclonic regimes govern circulation in the Eurasian Basin (red), with currents guided by local bathymetry, and following shelf-breaks and ridges (Carmack & Wassmann, 2006; Wassmann et al., 2015). Between the cyclonic and anticyclonic regimes of the two basins, the Trans-Polar Drift Stream (Orange) carries water from Siberia to the Fram Strait across the deep Arctic (Bluhm et al., 2015).

The NSR shown in Figure 1a predominantly crosses the Eurasian Arctic shelf. This region is comprised of five seas: the Barents, Kara, Laptev, East-Siberian, and Chukchi. The Barents and Chukchi seas are inflow shelves, and are influenced by incoming Atlantic and Pacific water, respectively (Carmack & Wassmann, 2006; Williams & Carmack, 2015). Atlantic water entering the Barents Sea Opening flows east across the shallow shelf. Dense water is formed in the Barents Sea, sinks, and becomes a branch of the Arctic Circumpolar Boundary Current (ACBC) (Aksenov et al., 2011). The ACBC has three cores: the main of which is a lower shelf slope current centered at around the temperature maximum in the Atlantic Water (AW) layer at ~300 m depth, originated from the AW inflow through Fram Strait, a deeper Barents Sea Branch at ca. 1,000 m, and as well as a surface/subsurface branch (down to 150 m depth in the water column), aka Arctic Shelf-Break Branch (ASBB) (Aksenov et al., 2011). We refer to all these as part of the ACBC. We assert that the above boundary current structure has been observed by the NABOS and other observational programs and has been simulated in high-resolution models.



At the other end of the Eurasian shelf, a smaller amount (\sim 1 Sv) of Pacific Water enters the Chukchi Sea via the Bering Strait (green in Figure 1b), where it can follow the Alaskan Shelf-Break Current toward Canada, flow into the East-Siberian Sea and potentially join the transpolar drift downstream, or enter the Beaufort Gyre (Aksenov et al., 2011; Carmack & Wassmann, 2006).

Between the Barents and Chukchi seas lie the interior seas of the Arctic (Kara, Laptev, and East-Siberian). These are influenced by freshwater runoff from the Ob and Yenisei near the coast, and from the ACBC along the shelf edge in the north (Carmack & Wassmann, 2006; Williams & Carmack, 2015). Wind forcing plays an important role here, with cyclonic summer atmospheric circulation favouring an eastward transport in the Laptev Sea. (Bauch et al., 2009). Significantly for communication with the deeper interior of the Arctic Ocean, Ekman pumping drives upwelling and downwelling at the shelf-break and associated cross-shelf exchange (Williams & Carmack, 2015).

While this sketch describes the general pattern of ocean circulation in the Arctic Ocean, the region is unsurprisingly also characterized by strong seasonal and interannual variability (in addition to the secular trend driven by anthropogenic climate change). All of these factors combine to make the Arctic Ocean a complex and variable region for understanding and planning spill responses.

In the following section, we outline the Lagrangian modeling technique used to investigate the impact of advection on a potential Arctic oil spill from various locations along the NSR (marked in Figure 1a). We then quantify this with various metrics, and discus the consequences of ocean circulation for the spread of pollutants from in the Arctic Ocean.

2. Methodology

2.1. NEMO (Nucleus for European Modeling of the Ocean)

In this study, we use the ORCA0083 1/12° resolution configuration of the NEMO (Nucleus for European Modelling of the Ocean) general circulation model (GCM) coupled to the Louvian-la-Neuve Ice Model (LIM2) sea-ice model (Fichefet & Maqueda, 1997; Goosse & Fichefet, 1999; Madec, 2014). Here we present model description relevant for this study, for more detail the reader is referred to Madec (2014).

NEMO is a global z-level model with a fully nonlinear free surface. Horizontal resolution in the Arctic is 3– 5 km, making it eddy-resolving in the central Arctic Ocean but only eddy-permitting on the shelves due to the small Rossby radius of deformation (Nurser & Bacon, 2014). The model has 75 vertical levels, with spacing varying from 1 m at the surface to 204 m at 6,000 m (there are 31 model levels in the upper 200 m). The model simulates vertical mixing using the turbulent kinetic energy (TKE) mixing scheme (Blanke & Delecluse, 1993) and uses the total variance dissipation (TVD) advection scheme for active tracers (Madec, 2014).

The LIM2 ice model uses Elastic-Viscous-Plastic rheology (EVP) (Hunke & Dukowicz, 1997), implemented on a C-grid (Bouillon et al., 2009), with thermodynamics based on two layers of ice and one layer of snow (Fichefet & Maqueda, 1997). It is coupled to the ocean model at every ocean time step, with a nonlinear quadratic drag of the sheer between the ice and ocean. The model has been found to accurately simulate sea-ice in the Arctic Ocean (Aksenov et al., 2017; Johnson et al., 2012; Wang et al., 2016).

The model is forced at the surface boundary using the DRAKKAR forcing set (DFS) atmospheric reanalysis (Brodeau et al., 2010). This is comprised of 6 hourly data for atmospheric winds (from the ERA40 reanalysis), temperature and humidity, daily radiative fluxes (shortwave and longwave) and monthly means for precipitation (rain and snow; from the CORE2 reanalysis) and runoff (Brodeau et al., 2010; Timmermann et al., 2005). In the simulation used here, NEMO was run from rest, with forcing from the beginning of 1978 until the end of 2015, and output saved as 5 day means.

NEMO is widely used by the research community for global studies at a variety of resolutions, including ORCA0083, (e.g., Duchez et al., 2014; Janout et al., 2015; Marzocchi et al., 2015; Srokosz et al., 2015). In the Arctic, it has been used extensively in the coarser 1/4° configuration (e.g., Aksenov et al., 2017; Lique et al., 2010; Popova et al., 2013). Evaluation of the circulation in NEMO (ORCA025), by way of calculating the baro-tropic stream function across the Arctic is presented in Lique et al. (2010), though the authors note the difficulty in accurately observing surface currents as sea surface height cannot be directly observed (Lique et al., 2010)—although reanalysis products are available and used here.



Although the recent realization of ORCA0083-N06 used here has undergone extensive validation globally, it has not been comprehensively evaluated in the Arctic. Further evaluation of the 1/12° NEMO ORCA0083-N06 run is presented in section 3.1. Here we compare ice cover against satellite-derived reanalysis data. We also compare modeled sea surface height with satellite observations. This is used in conjunction with an analysis of the model's barotropic stream function to evaluate the simulated circulation.

As with all models, NEMO is not without its limitations. For example, the configuration of NEMO used in these experiments lacks tides and wave model. We do not expect these to have a significant effect on our results, but it does place restrictions on the conclusions that we can draw in some coastal areas where tides play a particularly significant role in the ocean dynamics (Luneva et al., 2015; Padman & Erofeeva, 2004). Specifically, tides and waves are both important for mixing oil into the water column and dispersing it. However, given that we are modeling advective pathways—rather than directly modeling the physics of the oil itself, which would vary strongly based on the type of oil spilled—mixing is not directly accounted for anyway. We instead aim to describe the circulation pathways followed by the ocean currents in order to give a more general overview of where spilled oil could go, assuming it has already become mixed into the water column.

2.2. Lagrangian Modeling

There are two approaches by which ocean models can address pollutant dispersal: (1) online representation via a passive tracer, whose concentration is determined by the resolved circulation and the parameterized mixing—for more detail, see Madec (2014); and (2) offline, using saved output from a preexisting run of the model. This approach uses Lagrangian "particles," whose positions in space and time are updated according to the saved mean velocities and does not require the full model to be rerun (Blanke & Raynaud, 1997). Both have advantages and drawbacks.

The transport of online tracers is consistent with model transport processes of advection and diffusion, and mixed layer processes such as convection. NEMO employs Eulerian meshes to solve the tracer evolution equation numerically, thus this approach is often called "Eulerian." However, this Eulerian approach comes at a significant computational cost because it requires rerunning the high-resolution global model itself for each simulated spill scenario.

In contrast, by making use of output from an already existing (and computationally expensive) simulation, and by calculating for only a fixed number of trajectories, offline Lagrangian particles require significantly less computational resource. This reduced cost can particularly suit studies where trajectories are repetitively initialized from multiple sites at multiple time points, a situation which would require many separate simulations for online tracers. Using hundreds of thousands of discrete trajectories, it is possible using Lagrangian particles to identify advective pathways and their variations. The downsides of using this offline approach are that individual Lagrangian particles effectively represent large quantities of pollutant, and parameterized mixing processes are unrepresented, so we can only consider the effects of advection but not diffusion. This can be compensated for by using many particles over an ensemble of releases. Furthermore, while the Lagrangian method does allow for subduction due to nonzero vertical velocities, it cannot account for convection—i.e., vertical mixing due to water-column instability. This is most likely to affect the results in the areas of Atlantic inflow where convection is prevalent.

For the purposes of this study, we use the ARIANE particle-tracking software package which uses the Lagrangian methods outlined above to calculate the evolution of trajectories (Blanke & Raynaud, 1997). Here ARIANE reads in 3-D velocity fields from the 5 day mean NEMO output, and uses this to disperse virtual "particles" released into the model's flow field. These particles are transported per a bilinear interpolation of the velocity field, using an analytical method to solve for particle translation through model grid cells, and the resulting trajectories are stored for analysis at daily frequency. Although ARIANE does not include horizontal mixing (which, in part, represents subgrid processes in the ocean model), the high resolution of NEMO used here accounts for most of the relevant transport explicitly. ARIANE is mass-conserving and powerful for describing the large-scale, long-term impact of advection. However, it does not account for turbulent mixing, and since it works with 5 day mean advection fields, it does not guarantee that particles will exactly follow constant density surfaces.

Individual trajectories from each simulated pollution event can be plotted to highlight the different pathways followed by each particle, with the distribution of particles after some given time indicating dispersal size, distance, and shape. Using multiple releases initially close in space and time can provide information



on the uncertainty associated with different spill sites and dispersal routes. Here we use the term "advective footprints" to describe the ensemble of trajectories from a given release site. Lagrangian analysis, specifically using ARIANE and the advective footprint approach advocated here, has previously been employed to model oil spills and other pollutants (Main et al., 2016; Robinson et al., 2017).

2.3. Experiment Design

In order to evaluate the impact of advection of a potential Arctic oil spill, we consider the parts of the ocean most at risk of an accident occurring. The Northern Sea Route (NSR) is a major shipping corridor running between the Barents Sea in the west and the Bering Strait in the east (Lee & Kim, 2015). It offers a connection between North West Europe and the Far East which, in the future, could be more economically viable than the normal shipping route via the Suez Canal (Liu & Kronbak, 2010; Schøyen & Bråthen, 2011). It offers a 40% reduction in distance compared to the Suez route (Schøyen & Bråthen, 2011), and this has the potential to cut sailing times by up to 10 days (Aksenov et al., 2017). Both the reduced time at sea, and associated saving in fuel, make this a potentially attractive route for shipping companies.

To model the impact of advection on an oil spill from the NSR, virtual Lagrangian particles were released into the NEMO flow field at 15 sites along the Northern Sea Route (see Figure 1a). These locations were selected to sample the length of the NSR, between Murmansk in the west and the Bering Strait in the east, via the Barents, Kara, Laptev, East-Siberian, and Chukchi Seas. The NSR is not strictly defined as a single corridor, and different definitions have been used in the previous literature (e.g., Aksenov et al., 2017; Lee & Kim, 2015). Here we take this into account by defining a "main" route via straits connecting each sea, as well as a more northern "alternate" route.

Release sites at Murmansk and the Bering Strait were selected as the start and end points of both routes. The straits along the "main" route are potentially high-risk locations due to variable sea-ice cover, shallow water and restricted room for manoeuvre, which could increase the risk of a shipping accident. A further five sites were taken at the midpoints between these straits to sample each of the Eurasian Arctic seas. Finally, the borders between each sea were sampled from our northward "alternate" route to complete the set of release locations. The locations of each release site are marked in Figure 1a.

At each of the 15 sites, 100 particles were seeded over a regular 10 km \times 10 km grid, chosen to represent an area covered by oil after several days of spreading. These releases were repeated every 10 days throughout the navigable season of the NSR (taken to be start of June to end of October), every year from 1990 to 2009, to take into account seasonal and interannual variability of the circulation. All releases from a given year were conducted as part of the same experiment, resulting in twenty experiments of 22,500 particles.

All trajectories were advected with the full 3-D velocity field for 2 years from their respective launches, with the particles' positions recorded daily. We consider this 2 year period to account for spills which are not recovered before the Arctic freezeup begins, and therefore persist in the Arctic for beyond one winter. Previous research has evaluated dispersal over shorter time periods (1 year) but this has been in regions with warmer waters and no sea-ice (Main et al., 2016).

It is important to note that these particles do not directly represent real particles of oil. With only 100 particles per release, each is potentially representative of a large quantity of pollutant (see section 3.3.6. on the sensitivity of this assumption). Also, the particles are modeled here as being neutrally buoyant, and represent oil that is dissolved (or emulsified) within seawater rather than that which is floating as a sea surface slick or whose density changes with time.

2.4. Transport Metrics and Advective Footprints

We begin by simply plotting trajectories from each release site in order to give a qualitative description of the pathways that they follow (see section 3.2.1). Additionally, we define quantitative metrics to describe the distances traveled by particles, the uncertainty associated with where they go, and the sensitivity to when a spill occurs:

First, we are interested in how far particles go. This is trivial to quantify, and we used two metrics: total distance traveled (the sum of all the distances traveled in each time step) and A-B distance traveled by particles from each release site. These distances were calculated after (a) 9 months, to represent the typical time between a spill occurring in the summer and the melt season beginning next spring, and (b) after 2 years, to assess the fate of irrecoverable oil. In order to better describe the spread of Lagrangian particles, we introduce the concept of "advective footprints." Footprints are defined as the area of ocean covered by Lagrangian particles some time T after release. We choose T = 9 months (270 days) to correspond the approximate time between an oil spill occurring and the beginning of the melt season next spring.

To calculate the area covered by particles, we define a grid and count the number of cells occupied during the 270th time step (i.e., only cells occupied in the 270th time step, and not counting cells passed through on their way). We use a coarsened version of the ORCA12 grid—10 cells in the *i* direction and 11 in the *j* direction. This was chosen to provide approximately regularly sized square cells, and of a size that typically produced continuously filled footprints. The "area of advective footprint" is then defined to be the sum of the areas of each occupied grid cell.

This is calculated for each release (so 15 releases for each of the 20 years studied) individually. This figure is then averaged over the 300 releases for each site to give an estimate of the area that would likely be affected by a spill from a particular location—i.e., a measure of the horizontal spreading of our particles. We then compare how this value varies with respect to both season and year of launch to assess the interannual and intraannual variability of our experiments.

Additionally, we consider the "envelope" of all of these footprints—i.e., the total area covered by all 30,000 particles from a given launch site. The size of this "envelope" is representative of the uncertainty associated with pathways from a given location. Spilled oil could go anywhere reached by the Lagrangian particles, but it will not necessarily follow every pathway: each trajectory represents one of many possible pathways. This metric gives an overview of the areas potentially at risk.

Finally, we investigate the likelihood of subduction and the pathways followed by subducted particles. Oil that is entrained deep into the water column would probably not be recoverable, therefore it is necessary to understand where there is an enhanced risk of this happening. We take 100 m as an approximation for the maximum mixed layer depth in the Arctic Ocean, and compare the fraction of particles below this threshold after 9 months. While this is deeper than the shallow mixed layer depth in the Arctic Ocean, this threshold was chosen to ensure that particles reaching these depths were definitely below the upper mixed layer regardless of where in the Arctic they sank below 100 m.

3. Results

3.1. Model Evaluation

The global NEMO model has previously been validated in 1°, 1/4°, and 1/12° resolution configurations (Marzocchi et al., 2015). Here we use the latest (as of 2017) run of 1/12° NEMO, although thorough evaluation of the Arctic at this resolution has only been done in the Kara and Laptev seas (Janout et al., 2015). However, Arctic circulation and exchanges have been extensively validated in the coarser 1/4° version, with the model reproducing observed features (Aksenov et al., 2016; Lique et al., 2010; Popova et al., 2013). Nonetheless, model evaluation is an ongoing endeavor, and here we focus on the performance of key model features that are most pertinent to the present study: ocean circulation and sea-ice cover.

3.1.1. Ocean Circulation

In order to evaluate the surface currents produced by the NEMO model, two variables are investigated: sea surface height (SSH) and the barotropic stream function (BSF).

Sea surface height is verified against satellite Dynamical Ocean Topography (DOT) inferred by the CPOM (Centre for Polar Observation and Modelling) Envisat (2003–2011) (Armitage et al., 2016). The SSH field is used as a pan-Arctic proxy for observed and modeled surface geostrophic currents. To compensate for the two data sets being measured relative to different reference level—a geoid for satellite data and a spherical reference SSH for the model—the anomalies rather than absolute SSH are compared and presented in Figure 2.

In both the satellite-derived and modeled data, the magnitude of the differences in sea surface height (SSH) between the highest and lowest parts of the Arctic is approximately 1 m on average between 2003 and 2011. Key features such as the raised Beaufort Gyre are present in the model results and consistent with the satellite data. This suggests that NEMO accurately simulates the large-scale geostrophic flow at the surface.



Journal of Geophysical Research: Oceans



Figure 2. Comparison of NEMO modeled and satellite observed (from Envisat) sea surface height (SSH), as a proxy for ocean circulation. The barotropic stream function calculated for the NEMO model (not shown) shows a similar pattern. Key features visible in both data sets are a higher SSH in the Amerasian basin compared to the Eurasian/North Atlantic. The doming of the Beaufort Gyre is also clear in both. Note that Envisat has missing data over the pole due to lack of satellite coverage.

The barotropic stream function was calculated for the average NEMO velocity fields (not shown, though the barotropic stream function for the 1/4° version is discussed thoroughly in other work, Lique et al., 2010). It shows a similar spatial pattern to the SSH for the same period. Because most of the stratification in the Arctic is confined to the upper 200 m, the ocean circulation at intermediate depth is largely barotropic (Aksenov et al., 2011; Pnyushkov et al., 2015), and supports the interpretation that the barotropic flow in the model is also qualitatively in agreement with the satellite-inferred data—see Figure 2. **3.1.2. Sea-Ice**

Further evaluation is provided by comparing modeled and reanalysis (from the National Snow and Ice Data Center (NSIDC)) sea-ice extent. This is presented in Figure 3. Two measures are used to analyze the horizontal extent of sea-ice cover: we compare the mean summer (June–September) ice extent in the NEMO and







NSIDC data sets (Figure 3a), and we look at how well NEMO reproduces the seasonal cycle of ice growth/ decline (Figure 3b). Both metrics are compared between 1900 and 2012 to cover the period of our Lagrangian experiments. Sea-ice extent is taken to be the area covered with ice concentration greater than 15%.

Generally, NEMO reproduces real ice conditions well. The current downward trend in summer sea-ice extent is clearly visible (Figure 3a) with minima in 2007 and 2012 minima reproduced. Figure 3b shows the monthly cycle of ice growth and decline, were NEMO is consistently within one standard deviation of the reanalysis data.

NEMO's seasonal cycle is also in good agreement with the NSIDC reanalysis data (Figure 3b). The total extent of the modeled ice cover is typically an overestimate, especially during the winter months, but it is accurate to within two standard deviations of NSIDC reanalysis data. Spatially, the ice cover in NEMO and NSIDC is reasonably similar, with sea-ice present/absent in approximately the same areas.

For a more detailed evaluation of NEMO/LIM2, the reader is referred to Wang et al. (2016).

3.2. Lagrangian Experiments

3.2.1. Arctic Circulation Pathways and Time Scales

In order to assess the advective pathways likely to be important for an Arctic oil spill, Lagrangian particles were released from 15 sites along the Northern Sea Route. Releases were performed every 10 days from May to October, from 1990 to 2009. Examples of trajectories from each site are presented in Figure 4, showing all releases from 2000. Note that reds (corresponding to the earlier parts of trajectories) are plotted over yellows/greens/blues from later in the particles' journeys to highlight the fastest advective time scales.

From Figure 4, the first result that is immediately apparent is that, depending on where the particles were released 1 year can be sufficient to reach the center of the Arctic Ocean. Within 2 years it is possible for them to leave the Arctic completely and flow into the North Atlantic. The spatial scale of these advective pathways confirms that a pan-Arctic outlook is necessary to understand the long-term fate of potential oil spills from the NSR.

As is also clear from Figure 4, advective pathways in the Arctic are highly variable depending on where (and when) Lagrangian "particles" are released. First, we address the sensitivity to release site. Below we detail the spread of the particles through different regions of the Arctic Ocean and Arctic seas:

3.2.1.1. Barents and Kara Releases

The north-eastward flow of the Atlantic Water through the Barents Sea, the Barents Sea Branch (Aksenov et al., 2010, 2011), is the main circulation feature, and Lagrangian particles released in the Barents Sea show this pathway (see Figures 4a and 4b). The particles followed two main trajectories; (1) eastward flow within the Murmansk Current, then through the Kara Gate into the Kara Sea, and (2) flow northward of Novaya Zemlya toward the Eurasian shelf-break within the West Novaya Zemlya Current, where they flow via the St. Anna Trough and across the northern Barents Sea shelf into the deep Arctic Ocean (Aksenov et al., 2010). Trajectories are relatively fast in this part of the Arctic, with 4–5 month transient times to cross from Murmansk to the Kara Gate. The trajectories also map the cyclonic gyre in the Central Basin of the Barents Sea, with some of the particles entrained in the gyre and following the westward flow toward the Barents Sea Opening (cf. Figure 4a and Aksenov et al., 2010).

As in the Barents, northward and eastward currents dominate in the Kara Sea (Figures 4c–4e). The particles released in the Kara Sea travel via the St. Anna Trough in the Nansen and Amundsen basins of the Arctic Ocean (Dmitrenko et al., 2015), and also via the Vilkitsky Strait Current into the Vilkitsky Strait and the western Laptev Sea (Janout et al., 2015). There is an episodic flow back from the Kara Sea through the Kara Gate into the Barents Sea. This is the Litke Current, observed on some occasions (Pfirman et al., 1997), and found to be correlated with an atmospheric sea level pressure gradient across Novaya Zemlya between the Barents and Kara Seas, and by seasonal variations in the buoyant eastward current at the southern end of the Kara Gate.

After leaving the St. Anna Trough, particles go on to follow the Eurasian shelf-break within the Shelf-Break Current (SBC). This is one of the cores of eastward flow, guided by topography Arctic Circumpolar Boundary Current (ACBC; Aksenov et al., 2011), and along the rest of their trajectories subduction down the shelf slope was seen to occur (Figures 4c–4e). From there, particles recirculate in the Nansen and Amundsen basins (Figures 4e and 4f), eventually heading to exit the Arctic Ocean via the western part of Fram Strait toward the Atlantic. Very few particles released in the Kara Sea reach the Atlantic within the 2 years studied.





Figure 4. Advective pathways from each of the 15 release sites, showing all releases from every release in the year 2000 (arbitrary choice as an illustrative example). Release sites are: (a) Murmansk, (b) Barents Sea, (c) Kara Gate, (d) Kara Sea, (e) Novaya Zemlya, (f) Severnaya Zemlya, (g) Vilkitsky Strait, (h) Laptev Sea, (i) New Siberian Islands, (j) Sannikov Strait, (k) East-Siberian Sea, (l) Wrangel Island, (m) De Long Strait, (n) Chukchi Sea, (o) Bering Strait.

3.2.1.2. Laptev, East-Siberian and Chukchi Releases

Lagrangian particles released on the Laptev Sea shelf depict the well-established circulation pattern: they flow toward the shelf slope, where they can follow one of two routes: the ACBC or the Transpolar Drift Current (hereafter Transpolar Drift) from Siberia to the Fram Strait (Aksenov et al., 2011; Dmitrenko et al., 2005; Janout et al., 2015; Newton et al., 2008) (see Figures 4f–4i). Trajectories following the ACBC are strongly guided by the seabed topography of the Arctic, and branch away from the shelf-break at the Lomonosov Ridge (Woodgate et al., 2001). Here they join the Transpolar Drift and flow toward the Atlantic. Particles released from all sites around the Laptev Sea are able to reach the Atlantic Ocean within 2 years.

Trajectories in the East-Siberian Sea are notably slower than those in other shelf seas (see Figure 4I in particular). Currents in the area vary in time and space, and the particles are carried in all directions, owing to oppositely directed inflowing waters from the Laptev Sea on one side (which carries particles east when it dominates) and the Chukchi on the other (which carries particles west) (Timmermans et al., 2014). Particles from the East-Siberian Sea remain on the shelf for a considerable time, typically beyond a year. Once these trajectories enter the central Arctic Basin, the Transpolar Drift is the principal pathway.



Circulation in the Chukchi Sea is driven by inflow from the Pacific and by the wind (Aksenov et al., 2016; Timmermans et al., 2014). The majority of particles in our experiments follow Pacific Water pathways, of which three main routes are noted: (1) the western route into the Chukchi Sea through the Herald Canyon, (2) the eastern route via the shelf-break jet through the Barrow Canyon and along the Alaskan Shelf-break (Appen & Pickart, 2012) and (3) the route across the shelf crossing the Herald Shoal and the Hanna Shoal via the Central Channel (Timmermans et al., 2014). Trajectories originating in the Chukchi Sea can become entrained into the Beaufort Gyre within 2 years. This is illustrated in Figures 4m–4o. Further detail of the variability of these pathways is discussed in section 3.2.5.

3.2.2. Distance Traveled

As a first-order metric, we are interested in how far particles go. The mean distances (across all experiments) traveled by particles from each launch site is presented below in Table 1. The distances traveled after 9 months and after 2 years are presented, both as direct A-B and the total length of the path traveled for each of the 15 release sites used, listed from west to east. The full path is a proxy for average current speed along a trajectory, whereas the A-B distance represents how far away from the initial spill site a recovery operation would need to consider.

From Table 1, the distance traveled shows only modest sensitivity release site when the full path length is considered. All sites fall within 300 km of the mean distance traveled (1,497 km) across all sites for the first 9 months of advection: i.e., particles are transported at approximately the same speed regardless where they are seeded from.

However, this is not the case when considering the direct A to B distance (calculated as the shortest distance along the surface of the Fischer Spheroid) for each site. Excluding the Bering Strait, particles starting from release sites at the west of the Northern Sea Route traveled significantly further (typically about twice as far) as those from the eastern end of the NSR (excluding the Bering Strait site, which was more in line with the western sites).

Here the cutoff between west and east was the boundary between the Laptev and East-Siberian Seas. Particles originating from the Barents, Kara and Laptev seas traveled on average 771 km in nine months and 1,454 km in 2 years, whereas those from the East-Siberian Sea (including New Siberian Islands and Sannikov Strait release sites) and Chukchi sea (excluding Baring Strait site) traveled an average of 382 km in nine months and 834 km in 2 years.

Table 1

Average Distances Traveled by Particles From Each Release Site (and Standard Deviations)

	Distance in	9 months	Distance in 2 years		
Site	Full path (km) (S.D. (km))	A to B (km) (S.D. (km))	Full path (km) (S.D. (km))	A to B (km) (S.D. (km))	
1. Murmansk	1,641 (374)	826 (290)	3,333 (552)	1,549 (820)	
2. Barents Sea	1,440 (522)	709 (346)	2,943 (667)	1,662 (882)	
3. Kara Gate	1,490 (346)	537 (354)	3,228 (479)	1,559 (848)	
4. Novaya Zemlya	1,273 (388)	1,062 (456)	2,814 (710)	1,551 (543)	
5. Kara Sea	1,450 (372)	713 (527)	3,074 (502)	1,420 (535)	
6. Severnaya Zemlya	1,534 (404)	685 (277)	3,046 (586)	1,081 (984)	
7. Vilkitsky Strait	1,701 (394)	883 (292)	3,409 (536)	1,212 (506)	
8. Laptev Sea	1,463 (428)	754 (229)	3,103 (623)	1,661 (802)	
9. New Siberian Islands	1,249 (299)	491 (248)	2,824 (658)	1,351 (1086)	
10. Sannikov Strait	1,631 (369)	375 (221)	3,364 (515)	906 (806)	
11. East-Siberian Sea	1,721 (439)	327 (179)	3,349 (671)	484 (293)	
12. Wrangel Island	1,312 (418)	341 (198)	2,850 (658)	786 (506)	
13. De Long Strait	1,754 (425)	333 (159)	3,429 (421)	613 (356)	
14. Chukchi Sea	1,336 (346)	425 (215)	2,957 (520)	874 (418)	
15. Bering Strait	1,450 (431)	960 (201)	2,966 (563)	1,629 (382)	

Note. Two metrics are used here: "Full path": adding up all the distances traveled in all time steps, and "A to B": the straight-line distance between where the particle started and where it ended up after 9 months/2 years.



The marked difference in A-B distance despite the similar total path distance demonstrates that particles launched toward the west of the route (and also in the Bering Strait) follow more direct routes, whereas those in the east are more prone to recirculation.

3.2.3. Advective Footprints: Horizontal Spread and Uncertainty

We now characterize the horizontal spread of particles. This is done by dividing the ocean into a grid (full details in section 2.5) and counting the total area of occupied grid cells. Here we consider grid cells that are occupied 9 months after particles were initially released. Again, 9 months was chosen to represent a typical time between a spill occurring and the next melt season beginning. In the context of an oil spill, this corresponds to the area that should be expected to be reached by potentially contaminated waters.

Each release of 100 particles per release site is considered separately (so a total of 300 releases per site: 20 years of releases with 15 releases per year) and these are averaged to give a typical release footprint size for each release site. Particles were initially distributed over an area of 100 km². These footprint sizes are presented below in Table 2.

Table 2 shows that the spreading of particles is strongly dependent on where they are initially released from. There is a clear east/west divide, with western release sites (1–5) typically producing larger footprints and central (6–11) and eastern (12–15) ones producing smaller footprints.

The smallest footprints were associated from the middle of the Northern Sea Route, around Siberia in the Laptev and East-Siberian Seas (including the New Siberian Islands and Sannikov Strait release sites between them). The East-Siberian Sea site produced anomalously small footprints, on average less than 10,000 km². The average area for the other three sites in this group was 27,267 km² (average standard deviation 15,518 km²), making them only half the size of the average footprint area for the remaining 11 release sites.

The five westernmost release sites (Murmansk—Kara Sea) produced the five largest footprints. The mean area of footprints from these sites was 67,866 km² (average standard deviation 10,665 km²); 48% bigger than the average for the other release sites, excluding the anomalously small footprint from the East-Siberian Sea release site. Including all sites between Murmansk and the Vilkitsky Strait, the average area of footprints from the west of the NSR was 61,361 km² (average standard deviation 12,317 km²).

Considering each of the 300 releases from each of the 15 releases sites, the average area of advective footprints 9 months after particles were releases was 45,762 km² (average standard deviation 13,307 km²). Excluding the five largest and four smallest sites highlighted, the typical area of a footprint was 44,161 km² (average standard deviation 15,233 km²).

Table 2

Mean and Standard Deviation of the Total Area of "Advective Footprints" by Release Site

Site	(a) Footprint area after 9 months (km ²) (S.D. (km ²))		(b) Area of "envelope" of all footprints (km ²)	Ratio (b/a)
1. Murmansk	67,239	(9,905)	796,824	11.9
2. Barents Sea	66,083	(10,167)	772,741	11.7
3. Kara Gate	74,411	(8,217)	767,301	10.3
4. Novaya Zemlya	52,989	(15,403)	895,653	16.9
5. Kara Sea	78,608	(9,635)	871,982	11.1
6. Severnaya Zemlya	42,367	(18,373)	884,118	20.9
7. Vilkitsky Strait	47,606	(14,519)	767,249	16.1
8. Laptev Sea	25,028	(15,487)	644,895	25.8
9. New Siberian Isl.	30,650	(15,208)	545,667	21.8
10. Sannikov Strait	26,122	(15,860)	562,873	21.5
11. East-Siberian Sea	9,154	(8,293)	317,715	34.7
12. Wrangel Island	30, 474	(15,750)	586,770	19.3
13. De Long Strait	40,346	(14,567)	472,674	14.5
14. Chukchi Sea	48,613	(14,555)	593,230	12.2
15. Bering Strait	46,732	(13,663)	674,067	14.4
Average	45,762	(13,307)	676,917	14.8

Note. (a) These correspond to the total area of grid cells occupied by Lagrangian particles 9 months after they were initially released. Three hundred 300 distinct releases (20 years, 15 releases per year) were used to calculate these figures. (b) The same metric, but considering the "envelope" around footprints from all 300 experiments. (c) Ratio of a/b.



Additionally, the area of the "envelope" surrounding all 300 footprints from each site (again after 9 months) was calculated, i.e., the total area affected by at least one release. While the first metric addresses how waters from a particular release spread, this metric addresses part of the uncertainty in where they will go. As with the individual footprints, western release sites were found to have the largest "envelopes" while the smallest were in and around the East-Siberian Sea. The large footprint "envelopes" in the west were typically associated with particles being rapidly entrained into a well-organized current (the ACBC), which enabled them to travel further (see Table 1) and hence produce a larger envelope than particles which stay closer to their release sites.

Finally, we calculate the ratio of the "individual footprints" and the "envelopes." This demonstrates that the uncertainty in where spilled oil could go (the envelope) is an order of magnitude larger than the area over which contaminated waters would be expected to spread (the individual footprints). This ratio provides a measure of the variability of advective pathways from a given site: larger ratios correspond to footprints with less overlap between experiments, suggesting more variable circulation. The combination of the especially large envelope and ratio from the Severnaya Zemlya site is partly due to the highly variable circulation around this area—discussed further in section 3.2.5. The largest ratio was produced by the East-Siberian Sea releases, but it should be noted that this was due to both an abnormally small envelope and individual footprints, which stay relatively close to their initial release site.

3.2.4. Subduction

Having addressed the horizontal spread of our particles, we now look in the vertical direction. It is important to note that we are only considering subduction due to advection, as our Lagrangian technique does not explicitly include convection. All particles were initially released at the ocean surface, but they were not constrained to stay there. Here we investigate the fraction of particles which sink below 100 m depth. This was chosen as a rough approximation for the depth of the upper mixed layer, as any oil entrained deeper into the water column will be especially difficult to recover/unrecoverable.

Two snapshots were investigated: the time step after 9 months of advection and the time step after 2 years of advection, as with the distance traveled metrics. At these snapshots, particles were classified as either above or below a 100 m threshold. The results are presented below in Table 3.

From Table 3, it is clear that subduction is not a major concern for the majority of releases locations, with two notable exceptions: the Barents Sea and Novaya Zemlya sites (and, to a lesser extent, Murmansk and

Table 3

Percentages of Particles From Each Release Site Which are Below a 100 m Depth Threshold During the 270th Time Step of the Experiment (9 Months) and During the 730th (2 Years) for Each Release Site

	% of Trajectoriesdeeper than 100 m			
Site	After 9 months	After 2 vears		
1. Murmansk	8.01	17.67		
2. Barents Sea	18.36	28.68		
3. Kara Gate	1.19	5.80		
4. Novaya Zemlya	22.63	38.97		
5. Kara Sea	0.34	2.10		
6. Severnaya Zemlya	1.25	5.07		
7. Vilkitsky Strait	0.05	0.65		
8. Laptev Sea	n/a	0.49		
9. New Siberian Islands	n/a	0.16		
10. Sannikov Strait	n/a	0.01		
11. East-Siberian Sea	n/a	0.14		
12. Wrangel Island	0.13	1.00		
13. De Long Strait	0.25	2.25		
14. Chukchi Sea	2.48	7.58		
15. Bering Strait	5.61	9.71		

Note. The history of trajectories is not considered here, only whether they are above or below the threshold for the two snapshots studied.

the Bering Strait). One in five particles seeded from the first two sites ends up below the 100 m threshold after 9 months of advection, and is potentially unrecoverable. No particles from any release in any year were below 100 m after 9

months from four central release sites: these were in the Laptev and East-Siberian Seas, as well as the two release points on the border between these seas. We then investigated where this subduction occurs. An illustrative example, with trajectories colored to indicate their depth, is presented in Figure 5. In this figure, trajectories from 2007 are shown in three different groups, to highlight the regional differences in subduction.

From Figure 5, we can see that subduction mainly occurs for the western group of release sites, as noted in Table 3. It is apparent that this occurs mostly around the Eurasian shelf-break, where particles downwell across the shelf slope. These deep trajectories are guided by the bathymetry of the Arctic Ocean and tend to follow the Arctic Circumpolar Boundary current along the Eurasian shelf-break.

3.2.5. Sensitivity to Time of Release

The distance traveled, "advective footprints" and subduction metrics discussed in the previous sections were compared with respect to launch site. Here we repeat that analysis, but instead of comparing different release locations, we compare how these metrics vary with respect to time of release. This is presented in Table 4: first we compare the sensitivity to year of release (averaging over all release sites





Figure 5. Trajectories from a typical year (2007) plotted to highlight their depth. Trajectories run for two full years and are broken down into western, central and eastern groups of release sites to make the pathways more obvious. Deeper trajectories (purples) are plotted on top of shallower ones to highlight the pathways followed by subducted particles. The release sites used in each figure are denoted by yellow dots.

and all releases within the given year) and then across each of the 15 releases per year (averaging over all release sites and all years). For each of the three metrics, we look at a snapshot 9 months after particles were initially released.

From Table 4, it is clear that none of these metrics show significant sensitivity to year of release. Interannual variability is present, but with no clear trend in any of the metrics used here.

However, two of the metrics showed significant intraannual variability. The mean footprint area metric suggests slightly reduced spreading of particles correlating with later releases. The most notable trend came in the subduction metric. Here the fraction of particles subducted below 100 m steadily increases toward the end of the season, and it more than double between the first set of releases in June and the last set of releases in October.

As noted in section 3.2.5, subduction is not a major occurrence for most release sites. However, for those where it does matter, it is more likely to happen for particles launched in the autumn than those starting in the spring.

Investigating the sensitivity to time of release highlighted variability in the pathways that trajectories follow. Multiple major pathways exist for some sites, and three examples (specifically the Bering Strait, Kara Gate, and Severnaya Zemlya releases) with clear contrasts are presented in Figure 6.

Figure 6 highlights some of the variability of pathways discussed in section 3.2.1. Figures 6a and 6b show particles which were only launched 20 days apart from each other, yet flow in markedly different directions. Figure 6a highlights the Alaskan Shelf-Break current, which leads to trajectories along the North American coastline, before seeming to join the Beaufort Gyre. Figure 6b shows the other extreme, with particles flowing into the Chukchi Sea and affecting the Russian coastline.

Figures 6c and 6d show a more subtle variation. As previously noted, the dominant flow in the Kara Gate is from the Barents into the Kara Sea, but occasionally this is reversed. An example (Figure 6c) which shows this "reversed" flow west through the Kara Gate and then north-eastward along the Novaya Zemlya coast is shown. Figure 6d shows the usual eastward-only flow. This could be indicative of a wind-driven blocking event, analogous to similar events in the Vilkit-sky Strait (Janout et al., 2015). Further investigation (not shown) suggests that the usually dominant buoyancy driven current (west to

east) interplays with a wind-driven current in the opposite direction, which correlates with increased atmospheric pressure to the north-west of Novaya Zemlya.

Figures 6e and 6f compare consecutive releases from the Severnaya Zemlya site. Despite being launched just 10 days apart, they flow in opposite directions: all travel toward the Atlantic Ocean via the Fram Strait in Figure 6e, and all bar one follow the Arctic Circumpolar Boundary Current in Figure 6f. Both pathways are seen frequently, as this marks a region of variable flow: it is the boundary between where the eastward flow dominates in the Barents and Kara seas, and the east where the Transpolar Drift Stream is more common. **3.2.6. Sensitivity to Initial Area of Release**

So far in this study, we have used 100 km² as the initial area covered by our Lagrangian "particles." This was chosen as a first-order approximation for the size of a typical oil spill. We have investigated the spread of these particles and how this varies depending on when and where particles are released, but it is also important to assess how our choice of initial conditions could affect these results.



Table 4

Sensitivity to Year and Season of Release

	Mean	Mean A-B	% of Particles	
	footprint	distance	below	
Year	area (km²)	(km)	100 m depth	
Sensitivity to Year of R	elease			
1990	48,322	629	5.43	
1991	48,975	706	3.77	
1992	46,964	730	6.02	
1993	45,892	560	2.97	
1994	47,799	808	4.64	
1995	42,863	583	3.07	
1996	37,397	590	3.46	
1997	44,147	573	3.73	
1998	41,031	525	3.28	
1999	40,816	599	5.32	
2000	43,487	596	4.72	
2001	46,147	621	3.24	
2002	47,045	568	3.99	
2003	52,520	646	3.89	
2004	43,004	599	3.55	
2005	47,168	637	2.99	
2006	46,057	605	4.13	
2007	46,954	664	2.82	
2008	51,107	699	4.94	
2009	48,595	576	4.37	
Sensitivity to Season of	f Release			
Season				
June (early)	49,310	631	2.60	
June (mid)	49,021	647	2.99	
June (late)	49,842	634	2.62	
July (early)	49,349	632	3.30	
July (mid)	48,622	633	3.31	
July (late)	48,500	635	3.67	
August (early)	48,058	636	3.88	
August (mid)	47,032	621	3.92	
August (late)	45,461	612	4.64	
September (early)	44,787	602	4.53	
September (mid)	44,326	619	4.59	
September (late)	42,632	615	5.22	
October (early)	41,145	615	4.82	
October (mid)	38,769	618	5.11	
October (late)	38,392	623	5.36	

Note. The metrics described in the previous 3 section (area of footprint after 9 months, straight-line distance traveled in 9 months, and fraction of particles below 100 m deep after 9 months) are recalculated with respect to year and season of release.

In order to do this, we repeat our experiment for a typical year (2000 was chosen as it is in the middle of the period studied), but with particles released over different sized areas: 400 km² (20 km × 20 km), and 25 km² (5 km × 5 km). We keep the number of trajectories the same in each case. Exactly as before, we then calculate the area covered by particles 9 months after their release. We also compare this to the results from year 2000 in the standard 10 km × 10 km initial release experiments.

Differences in dispersion were apparent but small, with larger release areas leading to slightly larger distributions after nine months, see Table 5. Results from the 5 km \times 5 km, 10 km \times 10 km, and 20 km \times 20 km experiments are compared:

It is apparent from Table 5 that the sensitivity to size of initial release is low. Excluding the East-Siberian Sea release site, which had an anomalously low footprint area anyway, the experiments from the largest release boxes spread to an area only 30% larger than those from the smallest release boxes, despite the 800% increase in initial area.

Sensitivity to initial area of release was found to vary throughout the NSR. Particles launched from the west of the route (which had the largest footprints) showed less variability than those in the east. The seven western release sites (Murmansk—Laptev Sea) had an average increase of 12% between the largest and smallest release sites. This contrasted with a 48% average increase in the easternmost release sites (New Siberian Islands to Bering Strait, excluding the New Siberian Islands site). In short, the larger the footprint size, the more sensitivity to the size of the initial release box.

4. Discussion

Understanding the advective pathways in the Arctic Ocean is an essential prerequisite for understanding the long-term fate of oil (or other pollutants) that could be released into the Arctic Ocean. Although interest in using the Arctic Ocean for commercial shipping is already increasing, particularly along the Northern Sea Route (NSR) (Liu & Kronbak, 2010; Schøyen & Bråthen, 2011), the Arctic remains a unique logistical challenge due to its harsh environment and relative lack of infrastructure and marine services (Ho, 2010). This lack of infrastructure, coupled with the often-considerable distances to ports along the NSR mean that should an oil spill occur, cleaning up an oil spill in the Arctic is more challenging than in other parts of the world. Therefore, it is necessary to consider the risk of significant amounts of oil not being recovered before the winter freezeup begins.

Should this happen, oil is likely to remain an active pollutant for a number of years due to the slowed biodegradation in the cold Arctic waters (Fingas & Hollebone, 2003). This presents a serious risk for the polar ecosystem: the food web in the Arctic is short and therefore contamination of one species can strongly affect the whole ecosystem—trophic interactions in the Arctic are simple compared to in other parts of the world, so population changes in just one key species would have cascading effects throughout the whole ecosystem (Hop & Gjøsæter, 2013; Palumbi et al., 2008). Previous research has worked toward a quantitative risk assessment of Arctic oil spills, and the locations reached by spilled oil have been identified as key variables for understanding the ecological impacts of a potential spill (Nevalainen et al., 2017). This means that understanding the circulation patterns in the Arctic is of key importance, and it makes the advective footprints technique used here a particularly powerful tool for assessing the long-term impact of potential Arctic oil spills.





Figure 6. Comparison of changing advective pathways from 3 release sites. Bering Strait: (a) 23 July 2000 release, (b) 12 August 2000 release. Kara Gate: (c) 14 June 1993 release, (d) 3 August 1993 release. Severnaya Zemlya: (e) 14 July 2007 release, (f) 24 July 2007 release.

When considering the long-term fate of spilled oil, we need first to know how far it is likely to be transported during some specified time-frame. We took 9 months as the likely time between a spill occurring and it becoming accessible after the next melt season, and 2 years to provide a longer-term outlook, and then measured how far particles were from their start points at these intervals. Within 9 months, particles traveled a mean distance of 1,497 km from their start point (calculated along the full path). The straight-line distance showed considerable variation. A 628 km was the average for all release sites, but there was a clear east-west divide: particles from the west of the route (Barents, Kara, and Laptev) seas traveled almost twice as far as those in the east (East-Siberian and Chukchi Seas, excluding the Bering Strait): 771 km compared to 382 km. This suggests that the location of a spill is key for determining how far oil could be transported. After 2 years, particles were on average 1,223 km from their start point (484–1,662 km, depending on release location). Although the circulation in the Arctic is relatively slow compared to the rest of the world ocean, this demonstrates the first key result of this experiment: pan-Arctic modeling is required to understand the long-term (order of years) fate of spilled oil.

Having quantified the distance that spilled oil could travel, we move on to addressing the question of where it will go to and the uncertainty in how it will spread. To do this, we introduced the concept of advective footprints. This technique has recently been used in other parts of the world to study changing circulation patterns and potential impacts for environmentally sensitive areas (Polyakov et al., 2017; van Gennip



Table 5

Sensitivity to Initial Area of Release

Mean Area (km²) Covered After 9 Months From Initial Releases of Area (Standard Deviation)

Site	(a) Initial area 25 km ²		(b) Initial area 100 km ²		(c) Initial area 400 km ²		Ratio (c/a)
1. Murmansk	69,308	(10,195)	71,806	(6,444)	74,168	(5,802)	1.07
2. Barents Sea	53,635	(6,597)	56,641	(5,292)	58,901	(3,314)	1.10
3. Kara Gate	72,971	(7,671)	76,015	(6,990)	77,668	(4,086)	1.06
4. Novaya Zemlya	52,946	(15,838)	58,683	(15,427)	62,019	(15,757)	1.17
5. Kara Sea	79,577	(4,075)	84,184	(3,814)	85,905	(5,245)	1.08
6. Severnaya Zemlya	42,874	(18,022)	47,437	(12,525)	49,782	(9,079)	1.16
7. Vilkitsky Strait	46,731	(15,402)	48,715	(13,303)	46,941	(8,831)	1.22
8. Laptev Sea	19,346	(4,017)	22,490	(7,709)	32,425	(6,961)	1.68
9. New Siberian Islands	16,926	(6,771)	22,283	(8,886)	30,360	(10,107)	1.79
10. Sannikov Strait	27,471	(13,064)	34,226	(12,320)	42,537	(11,083)	1.55
11. East-Siberian Sea	2,428	(855)	3,844	(929)	5,153	(2,075)	2.12
12. Wrangel Island	23,031	(11,852)	27,935	(10,009)	31,694	(7,512)	1.38
13. De Long Strait	19,832	(14,197)	22,434	(12,346)	26,519	(10,307)	1.34
14. Chukchi Sea	51,334	(9,808)	47,184	(12,439)	57,515	(8,332)	1.12
15. Bering Strait	39,599	(13,160)	49,856	(16,537)	60,357	(12,901)	1.52

Note. The footprint sizes (in 1,000 km²) after 9 months from release are compared for three different experiments in 2000, with initial releases of 25 km², 100 km² (as in the main experiments) and 400 km². The ratios between the largest and smallest initial releases are also presented.

et al., 2017). The trajectories from our Lagrangian experiments are in agreement with circulation pathways described in the previous literature (e.g., Aksenov et al., 2011; Carmack & Wassmann, 2006; Williams & Carmack, 2015). Known pathways are clearly reproduced, with key features such as the Arctic Circumpolar Boundary Current (ACBC) (Aksenov et al., 2011) clearly visible in trajectories launched from all along the Northern Sea Route. These, along with the Transpolar Drift Stream (Steele et al., 2004) and Pacific inflow (Aksenov et al., 2016) are the main large-scale features seen in our experiments. More detailed descriptions of specific pathways are presented in section 3.2.1.

We defined the metric "area of advective footprints" to quantify the uncertainty associated with these pathways. We divided the Arctic into a regular grid and counted the number of grid cells that were occupied by particles a set time after particles were released (full details in section 2.5). The sum of the areas of occupied grid cells was taken to be the area of the advective footprint. In the context of an oil spill, the size of a footprint from an individual experiment represents the spreading of the spill, while the area of the "envelope" of all footprints corresponds to the uncertainty in where it will go—and hence the area of ocean that would need to be searched to find it. Location of oil has been identified as a key variable for environmental risk assessments (Nevalainen et al., 2017), so it is necessary to quantify the uncertainty associated with where oil is likely to end up. Advective footprints provide a robust mechanism for doing that.

It was found that after 9 months, a typical footprint had an area of $45,762 \text{ km}^2$, with a standard deviation of $13,307 \text{ km}^2$, considering all release sites with an ensemble of particles initially distributed over an area of 100 km^2 . In terms of an oil spill, this figure corresponds to the area of ocean that would likely need to be affected in the spring following an unsuccessful clean-up from the previous year. This also highlights the areas likely to be at risk of potentially acute biological impact.

The size of footprints was found to be sensitive to release location, and using the results presented in section 3.2.2, we can define three main areas along the NSR, each with its own characteristic behavior: a western region (in the Barents and Kara Seas, release sites 1–7), a central region (Laptev Sea and the East-Siberian Sea, release sites 8–11), and an eastern region (in the Chukchi Sea and Bering Strait, release sites 12–15). The west of the NSR was associated with the largest footprints after 9 months—up to 78,608 km² (from the Kara Sea) and on average 61,361 km². The middle of the route was associated with the smallest footprints, on average 22,739 km². The eastern group of release sites had typical footprints of 41,541 km² after 9 months.



In addition to considering the spread from individual experiments, the area covered by all 30,000 trajectories from a given site (all years/seasons of release) was calculated. This metric, termed the area of the "envelope" of footprints represents the uncertainty in where the oil will go—i.e., the area of ocean that would need to be monitored the next spring in order to find the spilled oil. This demonstrates another important conclusion: the uncertainty in where oil could go is highly dependent on where a spill occurs. Particles from the west of the NSR tend to follow the Arctic Circumpolar Boundary Current (ACBC). They travel the farthest and are associated with the largest envelopes of footprints, and therefore a spill here would likely to require a large area (over 800,000 km²) of ocean to be searched. Particles from the Siberian and eastern groups of release sites did not travel as far, and were associated with smaller envelopes of footprints—especially in the case of the East-Siberian Sea, where the envelope was only 317,615 km²—less than half the size of that of the largest envelopes. In short, the impact of advection on an oil spill occurring in the east of the NSR is likely to be more predictable than for a spill occurring in the west.

In addition to horizontal advection, it is important to consider the risk of potential subduction of oil. After the Deepwater Horizons spill in the Gulf of Mexico, unrecovered oil became suspended in deep water layers before eventually settling on the ocean floor (Valentine et al., 2014). We considered potential surface spills from shipping accidents in this research, but if some of this oil becomes entrained into deeper water, it is also likely to be irrecoverable. Therefore, it is important to understand where subduction occurs. We considered the fraction of particles that had sunk below 100 m depth after 9 months and also after 2 years. This was found to be of greatest concern for particles launched from the west of the NSR, particularly for release sites in the Barents Sea (see section 3.2.4).

From the three release sites in the Barents Sea (Murmansk, Barents, and Novaya Zemlya), 16% of particles were below 100 m depth after 9 months, rising to 28% after 2 years. A negligible (<1%) fraction of particles from the central group of release sites (in the Laptev and East-Siberian Seas) experienced subduction. The eastern group showed only modest subduction, mostly from particles released in the Bering Strait: 6% of particles from this site were below 100 m after 9 months and 10% after 2 years. Subduction primarily occurred as downwelling at the Eurasian shelf slope. This is consistent with results from previous investigation into downwelling in the Arctic Ocean. (Shapiro et al., 2003).

4.1. Limitations and Future Work

We are limited by dependency on a single simulation with the NEMO model: though this is a leading edge high-resolution model, intercomparison work would provide added verification and estimates of model structural uncertainty. Limitations associated with this configuration of NEMO, such as its lack of tidal forcing, means that areas strongly affected by tides may be less well represented than the Arctic in general (Padman & Erofeeva, 2004). Furthermore, despite its 1/12° resolution, given the small internal Rossby radius of deformation in the Arctic (Nurser & Bacon, 2014), NEMO is only eddy-permitting rather than fully eddy-resolving in all of the Arctic. As ever, this means that higher resolution would resolve more physical processes and thus potentially provide better results.

This study is also limited in that we are only considering the portion of the oil which remains mixed into the water column and does not become trapped in sea-ice—as can happen when oil becomes encapsulated into growing ice (Afenyo et al., 2015). This can, in principle, be addressed with a similar study where the Lagrangian software used here is modified to track ice drift rather than ocean currents. However, realizing this would require significant modification of the particle-tracking software used here that is beyond the scope of this research. Additionally, oil would only be encapsulated in ice during the winter, and deciding when to follow ice or waters is itself a nontrivial question. This presents an interesting opportunity for further research, in which similar analysis using the concept of advective footprints could be put to use.

As discussed in section 3.2.6, interesting small-scale changes in current direction were seen from certain release sites in these experiments. Some can be attributed to processes described in previous literature (Janout et al., 2015) while others, such as a potential wind-driven blocking in the Kara Gate, provide scope for further investigation.

While this study has focused upon considering the long-term fate of spilled oil, this is not the only pollutant that is of potential interest. In principle, a similar approach could be used to assess the impact of advection on other Arctic pollutants. As with the encapsulated oil example, this may require modification of the



Lagrangian software to realize this. The Arctic Monitoring and Assessment Programme (AMAP) identified four main categories of pollutant which pose a threat to the Arctic: persistent organic pollutants (POPs), "chemicals of emerging concern" (including flame retardants and pesticides), heavy metals and radioactive waste (AMAP, 2015). Some of these (e.g., riverine pollutants and pollution from nuclear submarines) are well suited to Lagrangian modeling as they spread from an easily defined source. This approach is not applicable in cases where we cannot accurately predict the locations where they interact with the ocean.

5. Conclusions

Circulation pathways and associated time scales in the Arctic Ocean have been investigated, and impact of oceanic advection for potential oil spills from the Northern Sea Route (NSR) has been explored. It has been demonstrated that pan-Arctic consideration of an oil spill's fate becomes necessary after a relatively short (1–2 years) time scale. The circulation patterns simulated are in agreement with observations.

Three main regimes have been defined to describe different sections of the NSR: western, central, and eastern. These are predominantly controlled by Atlantic Water, interior shelf-sea dynamics, and Pacific Water, respectively. Spills occurring at the west of the NSR are likely to travel the furthest (on average, 771 km within 9 months) and have the largest uncertainties associated with their pathways. Spills from the eastern and central groups travel less far (382 km), though those in the east have more uncertainty associated with their pathways than those in the center.

"Advective footprints" were introduced to quantify the area of ocean that would likely be affected in spring following an unsuccessfully cleared up oil spill the previous season. On average, this figure is 45,762 km², but it is notably higher in the western section (61,361 km²) and notably lower in the central section (22,739 km²) of the NSR. The 'envelope' of these footprints was introduced to quantify the uncertainty in where the oil could go—and hence how much of the ocean would need to be searched/considered potentially at risk. For a typical Arctic oil spill, this area was 676,917 km².

Subduction of oil, potentially leading to unrecoverable pollution, was identified as a risk for oil spills in the Arctic, especially those occurring in the west of the NSR. It also poses a risk in the east, though to a lesser extent. Subduction is unlikely to be a concern in the Laptev or East-Siberian Seas.

This study has provided a broad overview of the circulation features that would be likely to play a significant role in the event of an oil spill from the Northern Sea Route. We have been able to provide a general description of the directions that spilled oil would likely to be carried in, and the uncertainties associated with different regions. This provides the groundwork for more focused case studies in the event of an actual spill occurring.

References

Afenyo, M., Veitch, B., & Khan, F. (2015). A state-of-the-art review of fate and transport of oil spills in open and ice-covered water. Ocean Engineering, 119, 233–248.

Aksenov, Y., Bacon, S., Coward, A. C., & Nurser, A. J. G. (2010). The North Atlantic inflow to the Arctic Ocean: High-resolution model study. Journal of Marine Systems, 79, 1–22.

- Aksenov, Y., Ivanov, V. V., Nurser, A. J. G., Bacon, S., Polyakov, I. V., Coward, A. C., et al. (2011). The Arctic circumpolar boundary current. Journal of Geophysical Research: Oceans, 116, C09017. https://doi.org/10.1029/2010JC006637
- Aksenov, Y., Karcher, M., Proshutinsky, A., Gerdes, R., de Cuevas, B., Golubeva, E., et al. (2016). Arctic pathways of Pacific Water: Arctic Ocean model intercomparison experiments. *Journal of Geophysical Research: Oceans, 121*, 27–59. https://doi.org/10.1002/2015JC011299
- Aksenov, Y., Popova, E. E., Yool, A., Nurser, A. J. G., Williams, T. D., Bertino, L., et al. (2017). On the future navigability of Arctic sea routes: High-resolution projections of the Arctic Ocean and sea ice. *Marine Policy*, *75*, 300–317.

AMAP (2015). Summary for policy-makers: Arctic pollution issues 2015. Oslo, Norway: Arctic Monitoring and Assessment Programme. Appen, W.-J. V., & Pickart, R. S. (2012). Two configurations of the western arctic shelfbreak current in summer. Journal of Physical Oceanography, 42, 329–351.

Armitage, T. W. K., Bacon, S., Ridout, A. L., Thomas, S. F., Aksenov, Y., & Wingham, D. J. (2016). Arctic sea surface height variability and change from satellite radar altimetry and GRACE, 2003–2014. *Journal of Geophysical Research: Oceans*, 121, 4303–4322. https://doi.org/ 10.1002/2015JC011579

Bacon, S., Marshall, A., Holliday, N. P., Aksenov, Y., & Dye, S. R. (2014). Seasonal variability of the East Greenland Coastal Current. Journal of Geophysical Research: Oceans, 119, 3967–3987. https://doi.org/10.1002/2013JC009279

Bauch, D., Dmitrenko, I. A., Wegner, C., Hölemann, J., Kirillov, S. A., Timokhov, L. A., et al. (2009). Exchange of Laptev Sea and Arctic Ocean halocline waters in response to atmospheric forcing. *Journal of Geophysical Research*, 114, C05008. https://doi.org/10.1029/2008JC005062

Björk, G., Jakobsson, M., Rudels, B., Swift, J. H., Anderson, L., Darby, D. A., et al. (2007). Bathymetry and deep-water exchange across the central Lomonosov Ridge at 88–89°N. *Deep Sea Research, Part I, 54*, 1197–1208.

Acknowledgments

The underpinning high-resolution NEMO simulation was performed using the ARCHER UK National Supercomputing Service (http://www. archer.ac.uk). ARIANE simulations were performed using the JASMIN data analysis environment (http://www. jasmin.ac.uk). Arctic dynamic topography/geostrophic currents data were provided by the Centre for Polar Observation and Modelling, University College London (www.cpom.ucl.ac.uk/ dynamic_topography) (Armitage et al., 2016). Y. Aksenov was supported by Safer Operations at Sea—Supported by Operational Simulations (SOS-SOS) NE/N017099/1 and from the NERC Program "The North Atlantic Climate System Integrated Study (ACSIS), NE/ N018044/1." A. Yool and E. Popova were supported by NERC grants GOBLIN NE/M006972/1 and SIRENA NE/M00693X/1, OCTOPUS NE/ L008106/1, and National Capability in Ocean Modelling. Finally, we acknowledge the FAMOS (http://web. whoi.edu/famos/) program for providing a framework for many fruitful discussions which enhanced this work. Finally, we would like to thank Bob Newton and an anonymous second reviewer for their helpful comments which greatly improved this work.



- Blanke, B., & Delecluse, P. (1993). Variability of the tropical Atlantic Ocean simulated by a general circulation model with two different mixed-layer physics. *Journal of Physical Oceanography*, 23, 1363–1388.
- Blanke, B., & Raynaud, S. (1997). Kinematics of the Pacific equatorial undercurrent: An Eulerian and Lagrangian approach from GCM results. *Journal of Physical Oceanography*, 27, 1038–1053.
- Bluhm, B. A., Kosobokova, K. N., & Carmack, E. C. (2015). A tale of two basins: An integrated physical and biological perspective of the deep Arctic Ocean. *Progress in Oceanography*, 139, 89–121.
- Boe, J., Hall, A., & Qu, X. (2009). September sea-ice cover in the Arctic Ocean projected to vanish by 2100. Nature Geoscience, 2, 341–343.
 Bouillon, S., Morales Maqueda, M. Á., Legat, V., & Fichefet, T. (2009). An elastic-viscous-plastic sea ice model formulated on Arakawa B and C grids. Ocean Modelling, 27(3–4), 174–184
- Brodeau, L., Barnier, B., Treguier, A.-M., Penduff, T., & Gulev, S. (2010). An ERA40-based atmospheric forcing for global ocean circulation models. *Ocean Modelling*, 31, 88–104.
- Carmack, E., & Wassmann, P. (2006). Food webs and physical-biological coupling on pan-Arctic shelves: Unifying concepts and comprehensive perspectives. *Progress in Oceanography*, 71, 446–477.
- Carson, R. T., Mitchell, R. C., Hanemann, M., Kopp, R. J., Presser, S., & Ruud, P. A. (2003). Contingent valuation and lost passive use: Damages from the Exxon Valdez oil spill. *Environmental and Resource Economics*, 25, 257–286.
- Dmitrenko, I. A., Rudels, B., Kirillov, S. A., Aksenov, Y. O., Lien, V. S., Ivanov, V. V., et al. (2015). Atlantic water flow into the Arctic Ocean through the St. Anna Trough in the northern Kara Sea. *Journal of Geophysical Research: Oceans*, 120, 5158–5178. https://doi.org/10. 1002/2015JC010804
- Dmitrenko, I. A., Tyshko, K. N., Kirillov, S. A., Eicken, H., Hölemann, J. A., & Kassens, H. (2005). Impact of flaw polynyas on the hydrography of the Laptev Sea. *Global and Planetary Change*, 48, 9–27.

Duchez, A., Hirschi, J. J.-M., Cunningham, S. A., Blaker, A. T., Bryden, H. L., Cuevas, B. D., et al. (2014). A new index for the Atlantic meridional overturning circulation at 26°N. *Journal of Climate*, 27, 6439–6455.

- Fichefet, T., & Maqueda, M. A. M. (1997). Sensitivity of a global sea ice model to the treatment of ice thermodynamics and dynamics. Journal of Geophysical Research, 102, 12609–12646.
- Fingas, M. F., & Hollebone, B. P. (2003). Review of behaviour of oil in freezing environments. Marine Pollution Bulletin, 47, 333–340.
- Goosse, H., & Fichefet, T. (1999). Importance of ice-ocean interactions for the global ocean circulation: A model study. Journal of Geophysical Research, 104, 23337–23355.
- Ho, J. (2010). The implications of Arctic sea ice decline on shipping. Marine Policy, 34, 713-715.
- Hop, H., & Gjøsæter, H. (2013). Polar cod (Boreogadus saida) and capelin (Mallotus villosus) as key species in marine food webs of the Arctic and the Barents Sea. Marine Biology Research, 9, 878–894.
- Hunke, E. C., & Dukowicz, J. K. (1997). An elastic-viscous-plastic model for sea ice dynamics. Journal of Physical Oceanography, 27, 1849– 1867.
- Ivanov, V. V., & Golovin, P. N. (2007). Observations and modeling of dense water cascading from the northwestern Laptev Sea shelf. Journal of Geophysical Research, 112, C09003. https://doi.org/10.1029/2006JC003882
- Ivanov, V., Maslov, P., Aksenov, Y., & Coward, A. (2015). Shelf-basin exchange in the Laptev Sea in the warming climate: A model study. Geophysical & Astrophysical Fluid Dynamics, 109, 254–280.
- Izumiyama, K., Uto, S., & Sakai, S. (2004). Prediction of oil-ice sandwich formation. International Journal of Offshore and Polar Engineering, 14(3), 169–175.
- Janout, M. A., Aksenov, Y., Hölemann, J. A., Rabe, B., Schauer, U., Polyakov, I. V., et al. (2015). Kara Sea freshwater transport through Vilkitsky Strait: Variability, forcing, and further pathways toward the western Arctic Ocean from a model and observations. *Journal of Geophysical Research: Oceans, 120,* 4925–4944. https://doi.org/10.1002/2014JC010635
- Johnson, M., Proshutinsky, A., Aksenov, Y., Nguyen, A. T., Lindsay, R., Haas, C., et al. (2012). Evaluation of Arctic sea ice thickness simulated by Arctic Ocean Model Intercomparison Project models. *Journal of Geophysical Research*, 117, C00D13. https://doi.org/10.1029/ 2011JC007257
- Lee, T., & Kim, H. J. (2015). Barriers of voyaging on the Northern Sea Route: A perspective from shipping Companies. *Marine Policy*, 62, 264–270.
- Lique, C., Treguier, A. M., Blanke, B., & Grima, N. (2010). On the origins of water masses exported along both sides of Greenland: A Lagrangian model analysis. *Journal of Geophysical Research*, 115, C05019. https://doi.org/10.1029/2009JC005316
- Liu, M., & Kronbak, J. (2010). The potential economic viability of using the Northern Sea Route (NSR) as an alternative route between Asia and Europe. *Journal of Transport Geography*, *18*, 434–444.
- Luneva, M. V., Aksenov, Y., Harle, J. D., & Holt, J. T. (2015). The effects of tides on the water mass mixing and sea ice in the Arctic Ocean. Journal of Geophysical Research: Oceans, 120, 6669–6699. https://doi.org/10.1002/2014JC010310
- Macfadyen, A., Watabayashi, G. Y., Barker, C. H., & Beegle-Krause, C. J. (2013). Tactical modeling of surface oil transport during the deepwater horizon spill response. In Y. Liu et al. (Eds.), *Monitoring and modeling the deepwater horizon oil spill: A record-breaking enterprise*. Washington, DC: American Geophysical Union.
- Madec, G. (2014). "NEMO Ocean Engine" (Draft Edition r5171) "NEMO ocean engine" (Draft edition r5171). Note du Pôle de modélisation (No. 27). Paris, France: Institut Pierre-Simon Laplace. ISSN: 1288–1619.
- Main, C. E., Yool, A., Holliday, N. P., Popova, E. E., Jones, D. O. B., & Ruhl, H. A. (2016). Simulating pathways of subsurface oil in the Faroe-Shetland Channel using an ocean general circulation model. *Marine Pollution Bulletin*, 114, 315–326.
- Mariano, A. J., Kourafalou, V. H., Srinivasan, A., Kang, H., Halliwell, G. R., Ryan, E. H., et al. (2011). On the modeling of the 2010 Gulf of Mexico Oil Spill. Dynamics of Atmospheres and Oceans, 52, 322–340.
- Marzocchi, A., Hirschi, J. J. M., Holliday, N. P., Cunningham, S. A., Blaker, A. T., & Coward, A. C. (2015). The North Atlantic subpolar circulation in an eddy-resolving global ocean model. *Journal of Marine Systems*, *142*, 126–143.
- Meier, W. N., Hovelsrud, G. K., Van Oort, B. E. H., Key, J. R., Kovacs, K. M., Michel, C., et al. (2014). Arctic sea ice in transformation: A review of recent observed changes and impacts on biology and human activity. *Reviews of Geophysics*, *52*, 185–217.
- Nevalainen, M., Helle, I., & Vanhatalo, J. (2017). Preparing for the unprecedented—Towards quantitative oil risk assessment in the Arctic marine areas. *Marine Pollution Bulletin*, 114, 90–101.
- Newton, R., Schlosser, P., Martinson, D. G., & Maslowski, W. (2008). Freshwater distribution in the Arctic Ocean: Simulation with a highresolution model and model-data comparison. *Journal of Geophysical Research*, 113, C05024. https://doi.org/10.1029/2007JC004111
- North, E. W. E. W., Adams, E. E. E., Schlag, Z. Z., Sherwood, C. R. C. R., He, R. R., Hyun, K. H. K. H., et al. (2013). Simulating oil droplet dispersal from the deepwater horizon spill with a Lagrangian approach. In Y. Liu et al. (Eds.), *Monitoring and modeling the deepwater hori*zon oil spill: A record-breaking enterprise. Washington, DC: American Geophysical Union.



Nurser, A. J. G., & Bacon, S. (2014). The Rossby radius in the Arctic Ocean. Ocean Science, 10, 967–975.

Østreng, W., Eger, K. M., Fløistad, B., Jøgensen-Dahl, A., Lothe, L., MejInder-Larsen, M., et al. (2013). Shipping in Arctic waters: A Comparison of the Northeast, Northwest and Trans Polar Passages. Berlin, Germany: Springer. Retrieved from https://link.springer.com/chapter/10. 1007/978-3-642-16790-4 8

Overland, J. E., & Wang, M. (2013). When will the summer Arctic be nearly sea ice free?. Geophysical Research Letters, 40, 2097–2101. https:// doi.org/10.1002/grl.50316

Padman, L., & Erofeeva, S. (2004). A barotropic inverse tidal model for the Arctic Ocean. Geophysical Research Letters, 31, L02303. https:// doi.org/10.1029/2003GL019003

Palumbi, S. R., Mcleod, K. L., & Grünbaum, D. (2008). Ecosystems in action: Lessons from marine ecology about recovery, resistance, and reversibility. *BioScience*, 58, 33–42.

Peterson, C. H., Rice, S. D., Short, J. W., Esler, D., Bodkin, J. L., Ballachey, B. E., et al. (2003). Long-term ecosystem response to the Exxon Valdez oil spill. Science, 302, 2082–2086.

Pfirman, S. L., Kögeler, J. W., & Rigor, I. (1997). Potential for rapid transport of contaminants from the Kara Sea. Science of the Total Environment, 202, 111–122.

Pnyushkov, A. V., Polyakov, I. V., Ivanov, V. V., Aksenov, Y., Coward, A. C., Janout, M., et al. (2015). Structure and variability of the boundary current in the Eurasian Basin of the Arctic Ocean. *Deep Sea Research, Part I*, 101, 80–97.

Pnyushkov, A. V., Polyakov, I. V., Ivanov, V. V., & Kikuchi, T. (2013). Structure of the Fram Strait branch of the boundary current in the Eurasian Basin of the Arctic Ocean. *Polar Science*, *7*, 53–71.

Polyakov, I. V., Pnyushkov, A. V., Alkire, M. B., Ashik, I. M., Baumann, T. M., Carmack, E. C., et al. (2017). Greater role for Atlantic inflows on sea-ice loss in the Eurasian Basin of the Arctic Ocean. *Science*, *356*(6335), 285–291.

Popova, E. E., Yool, A., Aksenov, Y., & Coward, A. C. (2013). Role of advection in Arctic Ocean lower trophic dynamics: A modeling perspective. Journal of Geophysical Research: Oceans, 118, 1571–1586. https://doi.org/10.1002/jgrc.20126

Proshutinsky, A., Dukhovskoy, D., Timmermans, M.-L., Krishfield, R., & Bamber, J. L. (2015). Arctic circulation regimes. *Philosophical Transactions of the Royal Society A*, 373(2052), 1–18. Retrieved from http://rsta.royalsocietypublishing.org/content/roypta/373/2052/20140160. full.pdf

Proshutinsky, A. Y., & Johnson, M. A. (1997). Two circulation regimes of the wind-driven Arctic Ocean. Journal of Geophysical Research: Oceans, 102, 12493–12514.

Robinson, J., New, A. L., Popova, E. E., Srokosz, M. A., & Yool, A. (2017). Far-field connectivity of the UK's four largest marine protected areas: Four of a kind? *Earth's Future*, 5, 475–494.

Schøyen, H., & Bråthen, S. (2011). The Northern Sea Route versus the Suez Canal: Cases from bulk shipping. Journal of Transport Geography, 19, 977–983.

Shapiro, G. I., Huthnance, J. M., & Ivanov, V. V. (2003). Dense water cascading off the continental shelf. *Journal of Geophysical Research*, 108(C12), 3390. https://doi.org/10.1029/2002JC001610

Srokosz, M. A., Robinson, J., Mcgrain, H., Popova, E. E., & Yool, A. (2015). Could the Madagascar bloom be fertilized by Madagascan iron? Journal of Geophysical Research: Oceans, 120, 5790–5803. https://doi.org/10.1002/2015JC011075

Steele, M., Morison, J., Ermold, W., Rigor, I., Ortmeyer, M., & Shimada, K. (2004). Circulation of summer Pacific halocline water in the Arctic Ocean. Journal of Geophysical Research, 109, C02027. https://doi.org/10.1029/2003JC002009

Timmermann, R., Goosse, H., Madec, G., Fichefet, T., Ethe, C., & Dulière, V. (2005). On the representation of high latitude processes in the ORCA-LIM global coupled sea ice-ocean model. *Ocean Modelling*, *8*, 175–201.

Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., Mccall, M., et al. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. *Journal of Geophysical Research: Oceans*, 119, 7523–7548. https://doi.org/10.1002/ 2014JC01027

Valentine, D. L., Fisher, G. B., Bagby, S. C., Nelson, R. K., Reddy, C. M., Sylva, S. P., et al. (2014). Fallout plume of submerged oil from Deepwater Horizon. Proceedings of the National Academy of Sciences of United States of America, 111, 15906–15911.

Van Gennip, S. J., Popova, E. E., Yool, A., Pecl, G. T., Hobday, A. J., & Sorte, C. J. B. (2017). Going with the flow: The role of ocean circulation in global marine ecosystems under a changing climate. *Global Change Biology*, *23*, 2602–2617.

Wang, Q., Ilicak, M., Gerdes, R., Drange, H., Aksenov, Y., Bailey, D. A., et al. (2016). An assessment of the Arctic Ocean in a suite of interannual CORE-II simulations. Part I: Sea ice and solid freshwater. Ocean Modelling, 99, 110–132.

Wang, M., & Overland, J. E. (2012). A sea ice free summer Arctic within 30 years: An update from CMIP5 models. *Geophysical Research Letters*, 39, L18501. https://doi.org/10.1029/2012GL052868

Wassmann, P., Kosobokova, K. N., Slagstad, D., Drinkwater, K. F., Hopcroft, R. R., Moore, S. E., et al. (2015). The contiguous domains of Arctic Ocean advection: Trails of life and death. Progress in Oceanography, 139, 42–65.

Weisberg, R. H., Lianyuan, Z., & Liu, Y. (2017). On the movement of Deepwater Horizon Oil to northern Gulf beaches. Ocean Modelling, 111, 81–97.

Williams, W. J., & Carmack, E. C. (2015). The 'interior' shelves of the Arctic Ocean: Physical oceanographic setting, climatology and effects of sea-ice retreat on cross-shelf exchange. *Progress in Oceanography*, 139, 24–41.

Woodgate, R. A., Aagaard, K., Muench, R. D., Gunn, J., Björk, G., Rudels, B., et al. (2001). The Arctic Ocean Boundary Current along the Eurasian slope and the adjacent Lomonosov Ridge: Water mass properties, transports and transformations from moored instruments. Deep Sea Research, Part I, 48, 1757–1792.

Xie, H., Yapa, P. D., & Nakata, K. (2007). Modeling emulsification after an oil spill in the sea. Journal of Marine Systems, 68, 489-506.